\nocite{Cunha.sep.61.213}
\nocite{Cunha.sep.65.113}
\nocite{Cunha.sep.65.167}
\nocite{Cunha.sep.67.31}
\nocite{Cunha.sep.67.107}
\nocite{Cunha.sep.70.193}
\nocite{Cunha.sep.70.227}
\nocite{Cunha.sep.70.293}
\nocite{Cunha.sep.72.83}
\nocite{Cunha.sep.72.101}
\nocite{Cunha.sep.72.111}
\nocite{Cunha.sep.72.181}
\nocite{Cunha.sep.73.21}
\nocite{Cunha.sep.73.243}
\nocite{Cunha.sep.73.317}
\nocite{Cunha.sepphd.74}


\title{Report bibliography}
This is my bibliography of articles
in the progress reports of the Stanford Exploration Project
written from 1974 to the present.
There are  15 
articles.
{\footnotesize
\putbib[SEP]
}
